Author’s response to reviews

Title: EFFECT OF VITAMIN D FORTIFIED FOODS ON BONE MARKERS AND MUSCLE STRENGTH IN WOMEN OF PAKISTANI AND DANISH ORIGIN LIVING IN DENMARK: A RANDOMISED CONTROLLED TRIAL

Authors:
Ida Grønborg (idamag@food.dtu.dk)
Inge Tetens (ite@nexs.ku.dk)
Elisabeth Andersen (elian@cancer.dk)
Michael Kristensen (mikr@kp.dk)
Rikke Larsen (rila@phmetropol.dk)
Thanh Tran (tllt@phmetropol.dk)
Rikke Andersen (rian@food.dtu.dk)

Version: 2 Date: 07 Nov 2019

Author’s response to reviews:

Reviewer reports:

Reviewer #1: The authors have revised the article perfectly, though I don't agree with their decision of removing of the phrases on ethnicity as while environment is of great importance and adding the time they lived in Denmark is quite necessary, ethnicity and genetic plays an important role on vitamin D and BTMs so I think it should be taken into account and so if the necessary information is not completely available, then just mention it as limitation.

Thank you for the comment on the revised manuscript. The information on the time the participants have lived in Denmark is available in Table 1 as a result of your previous comment. Was it something other than this that you were missing in the text?

You are referring to some removed phrases, unfortunately I am not sure which these are. We will be very willing to add information to the limitations section if necessary.